Author’s response to reviews

Title: Respiratory mechanics in infants with severe bronchiolitis on controlled mechanical ventilation

Authors:

Pablo Cruces (pcrucesr@gmail.com)
Sebastián Gonzalez-Dambrauskas (sgdambrauskas@gmail.com)
Julio Quilodrán (julio.quilodran@gmail.com)
Jorge Valenzuela (clarkinekit@yahoo.es)
Javier Martínez (javimartinez606@gmail.com)
Natalia Rivero (natyrivero.kine@gmail.com)
Pablo Arias (pro944@gmail.com)
FRANCO DIAZ (francodiazr@gmail.com)

Version: 5 Date: 14 Aug 2017

Author’s response to reviews:

1. Thank you for explaining why the raw data cannot be publicly shared. Please state why your data cannot be shared in the Availability of data and materials section.

   It was added

2. Please update your abstract in the submission system, as it is currently not the same as the abstract presented in your manuscript.

   it was uploaded.

3. In the Ethics approval section, please include the ethics committee reference numbers if available.

   No reference number is available.

   Thank you